Author's response to reviews

Title: Necrotizing anterior uveitis, anterior scleritis and ischaemic optic neuropathy in association with Streptococcal Toxic Shock Syndrome

Authors:

Konstantinos I Papageorgiou (papageorgiouk@doctors.org.uk)
Alexander S Ioannidis (alexioannidis@hotmail.com)
Petros S Andreou (pandreou@aol.com)
Ajay Sinha (Ajay.Sinha@meht.nhs.uk)

Version: 2 Date: 17 May 2007

Author’s response to reviews:

The title has been updated with the phrase case report
The abstract has been included
The abbreviations have been explained
The consent form statement has been added as well as the authors contribution